Author’s response to reviews

Title: A novel approach to neuraxial anesthesia: application of an automated ultrasound spinal landmark identification

Authors:
Ting Ting Oh (oh.ting.ting@singhealth.com.sg)
Mohammad Ikhsan (mohammad.ikhsan@u.nus.edu.sg)
Kok Kiong Tan (eletankk@nus.edu.sg)
Sultana Rehena (rehena.sultana@duke-nus.edu.sg)
Nian-Lin Reena Han (han.nianlin@kkh.com.sg)
Alex Tiong Heng Sia (alex.sia.t.h@singhealth.com.sg)
Ban Leong SNG (sng.ban.leong@singhealth.com.sg)

Version: 4 Date: 31 Mar 2019

Author’s response to reviews:

31 March 2019

To: Amelia De Salis / Amy Branch-Hollis
BMC Anesthesiology

Dear Amelia / Amy,

RE: “A novel approach to neuraxial anesthesia: application of an automated ultrasound spinal landmark identification” (Submission ID: BANE-D-18-00450R2)

Thank you very much for your message dated 29 March 2019.
Please find below point-by-point responses to each of the comments (in Italic red font). Amendments to the text are indicated in our responses.

-----------------------------------------------------------------------------------------------------

Editor Comments:

1. As a member of the editorial board (Associate Editor) of this journal, in order to ensure transparency, please declare this in the Competing Interests section of the Declarations.

-------- The Competing Interests section has been revised as: “Dr Ban Leong SNG is an associate editor of BMC Anesthesiology. The other authors declare that they have no competing interests.”

2. Please ensure that the title matches that which has been listed on our editorial system—currently they differ.

-------- The title has been changed accordingly.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

-------- The files had been uploaded according to the requirements.